

Author's response to reviews

Title: Effect of statin therapy on the progression of coronary atherosclerosis: new insights through meta-analysis

Authors:

Jinwei Tian PhD (tianjinweidr2009@163.com)

Xia Gu MD (shijikuihua@126.com)

Yun Xiao PhD (yunxiao@163.com)

Sining Hu MD (sininghudr@163.com)

Bo Yu PhD (yubodr2009@163.com)

Version: 2 **Date:** 20 December 2011

Author's response to reviews:

the manuscript is not newest.